Author's response to reviews

Title: Considerations for Increasing the Competencies and Capacities of the Public Health Workforce: Assessing the Training Needs of Public Health Workers in Texas

Authors:

Stephen Borders (stephen-b-borders@tamu.edu)
Craig Blakely (blakely@srhs.tamushsc.edu)
Barbara Quiram (quiram@srhs.tamushsc.edu)
Kenneth McLeroy (KMcLeroy@srhs.tamushsc.edu)

Version: 2 Date: 16 May 2006

Author's response to reviews:

I addressed the reviewers' suggestions - 1) updating the literature review 2) trying to give the article a broader perspective outside Texas and 3) making several syntax corrections.